

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dalgic, Basak; Fazlioglu, Burcu; Gasiorek, Michael

# Working Paper Costs of trade and self-selection into exporting and importing: The case of Turkish manufacturing firms

Economics Discussion Papers, No. 2015-17

**Provided in Cooperation with:** Kiel Institute for the World Economy – Leibniz Center for Research on Global Economic Challenges

*Suggested Citation:* Dalgic, Basak; Fazlioglu, Burcu; Gasiorek, Michael (2015) : Costs of trade and self-selection into exporting and importing: The case of Turkish manufacturing firms, Economics Discussion Papers, No. 2015-17, Kiel Institute for the World Economy (IfW), Kiel

This Version is available at: https://hdl.handle.net/10419/107775

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

http://creativecommons.org/licenses/by/3.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Discussion Paper No. 2015-17 | March 09, 2015 | http://www.economics-ejournal.org/economics/discussionpapers/2015-17

# **Costs of Trade and Self-selection into Exporting and Importing: The Case of Turkish Manufacturing Firms**

Basak Dalgic, Burcu Fazlioglu, and Michael Gasiorek

### Abstract

This paper focuses on self-selection into trade by exporting and importing firms, and on the presence of differential variable and sunk costs between exporters and importers across different categories of imports. In addition the authors consider the role of intensive and extensive margins with respect to products or countries. They use a rich and recent dataset for Turkish manufacturing firms for the period 2003–2010. This allows them to provide a comprehensive analysis of firm heterogeneity and the connection between firm-level performance and international trade. They provide evidence on the remarkable heterogeneity across firms where only-importers (importers) perform better than only-exporters (exporters). They detect a self-selection effect for both importing and exporting firms with a stronger effect for importers. The results suggest that the nature of sunk costs varies between importing and exporting activities with importers facing higher sunk costs. Tariffs represent a potentially important source of variation in the variable costs of trading. When taking the tariffs faced by firms into account, the authors find that the self-selection effect associated with sunk costs is still present but greatly reduced with a smaller reduction for importers compared to exporters.

(Published in Special Issue Micro-econometric Analyses of International Firm Activities)

**JEL** D24 F10 M20 L10

Keywords Firm heterogeneity; self-selection; sunk costs; exports; imports

#### Authors

Basak Dalgic, Hacettepe University, Ankara, Turkey Burcu Fazlioglu, Imes TOBB ETU University, Ankara, Turkey, bfazlioglu@etu.edu.tr Michael Gasiorek, University of Sussex, Brighton, United Kingdom

**Citation** Basak Dalgic, Burcu Fazlioglu, and Michael Gasiorek (2015). Costs of Trade and Self-selection into Exporting and Importing: The Case of Turkish Manufacturing Firms. Economics Discussion Papers, No 2015-17, Kiel Institute for the World Economy. http://www.economics-ejournal.org/economics/discussionpapers/2015-17

# 1 Introduction

This paper analyzes the existence of self selection mechanism associated with both exporting and importing activities of firms in Turkey with a special focus on whether a stronger self selection mechanism is at work for importing activities than exporting activities. In order to get a better understanding of the mechanics at play, we search for the possible heterogeneity of both sunk costs and variable costs across firms by trading status. While doing so we aim to expand the empirical evidence on firm heterogeneity in international trade by offering a comprehensive analysis of Turkish firms' international trading activities. We further handle exporting and importing activities of firms along with their diversification patterns.

The international trade literature has witnessed a dramatic change over the past eighteen years where the focus has switched from the investigation of macro level agents to micro players of trade. In this context, firm heterogeneity in international trade has become a core topic. The microeconometrics of firms' engagement in international trade was pioneered by Bernard and Jensen (1995), Aw and Hwang (1995) and Roberts and Tybout (1997). The theoretical framework underlying the literature has been largely stimulated by the seminal works of Melitz (2003) and Bernard et al. (2003). With the availability of firm level datasets a substantial empirical literature has shown that internationalized firms show superior performance to the firms who serve only the domestic markets<sup>1</sup>. The majority of the empirical literature exclusively focus on exports, with much less of a focus on imports. In particular, there are relatively few studies on the importing activity and firm-performance nexus for developing countries.

The big picture from this literature suggests that the superior performance of internationalized firms emerges via self-selection and post-entry effects. Regarding the latter, on the one hand firms become more efficient after they begin exporting through learning, or as a result of economies of scale via interaction with foreign clients, and being exposed to intensive competition in international markets. On the other hand, post-entry mechanisms of importing suggest a strong learning effect through importing intermediate and capital goods via international knowledge spillovers, variety effects and quality effects. The self selection hypothesis which emerges from the theoretical literature on export behavior of firms suggests that, due to the existence of sunk costs and different productivity levels within the same industry, only the most productive firms self-select into export markets. Similarly, the self-selection of more productive firms into import markets arises due to the fixed costs of importing. More recent literature on self-selection provides insights as to the possible heterogeneity of sunk costs and thus of self-selection mechanism

 $<sup>^1\</sup>mathrm{See}$  Greenaway and Kneller, 2007; Wagner, 2007, 2012 and Redding, 2010 for surveys of the empirical evidence.

across importing and exporting activities. While exporters face sunk costs linked to knowledge of markets, marketing and advertising, and the set-up of foreign distribution channels importers do not typically face these costs. Importers face greater informational asymmetries associated with imperfect monitoring of the quality of the imported goods, and the costs associated with transferring and utilising the embedded technology (Altomonte and Békés, 2009).

We utilize the most recent available dataset covering the whole population of Turkish manufacturing firms<sup>2</sup> with more than 19 employees matched with international trade data over the period 2003-2010. Being an emerging economy for whom trade has been an important driver of growth<sup>3</sup>, our case constitutes an interesting quasi-natural experiment since our data covers a period in which Turkey experienced a trade boom and underwent a structural transformation in terms of its production and trade patterns. The process of integration of the Turkish economy into the world economy gained momentum following the positive stimulus from the Customs Union with the EU in the late 1990s and the EU's decision to start accession talks with Turkey in 2004 accompanied by abundant foreign capital inflows. Further, following a series of macroeconomic and structural reforms, the Turkish economy recovered relatively quickly from the negative shock of the economic crisis in 2001. We analyze the period after 2002, over which Turkey experiences this recovery and a dramatic export boom  ${}^{4},{}^{5}$ . In the meantime, Turkey has undergone a structural transformation process both in terms of production and trade patterns along with sectoral and geographical diversification<sup>6</sup>.

<sup>&</sup>lt;sup>2</sup>Over the period the share of Turkish manufacturing industry in GDP was 23.5 percent on average. While manufacturing industry constituted 13.5 percent of overall employment in Turkey, it generated 93.5 percent of the total export volume. Although it has subsequently declined to around 80 percent of total exports, with this share Turkey is second to only China among the BRIC countries in terms of the share of manufacturing in exports. With such a large share the manufacturing industry plays an important role in determining Turkish export performance.

<sup>&</sup>lt;sup>3</sup>Turkey is an upper-middle income country who is the  $16^{th}$  in the World and  $6^{th}$  in Europe. It grew with an average annual real GDP growth rate of 5 percent over the past decade. As the GDP levels more than tripled to USD 786 billion in 2012, up from USD 231 billion in 2002, GDP per capita rised to USD 10,504, up from USD 3,500 between 2002 and 2012 whereas foreign trade volume constituted 49.5 percent of its GDP on average over the given period.

<sup>&</sup>lt;sup>4</sup>Turkey's total trade volume increased from \$88 billion in 2002 to \$389 in 2012, an increase of 342 percent in a decade's time. Turkey's exports increased by 325 percent (to \$153 billion from \$36 billion) over the same period. This compares to the average export performance of its peers in the same income group (Brazil, China, Mexico, and South Africa) whose exports grew by 212 percent in the same period.

<sup>&</sup>lt;sup>5</sup>Due to the global financial crisis, being an open and free market economy Turkey was adversely affected by the declining external demand and falling international capital flows similarly to other emerging markets. Turkey experienced a decline in its export volume by 33 percent between 2008 an 2009

<sup>&</sup>lt;sup>6</sup>2002-2012 period witnesses a structural shift away from traditional export sectors of textiles and clothing towards machinery and metals. A transition across destination markets occurs

We add four main contributions to the literature on trade and firm heterogeneity. First of all, to the best of our knowledge our paper is the first attempt to investigate self-selection mechanisms for Turkey. There are limited number of studies that simultaneously analyze import and exporting behavior with even more rare evidence on self-selection of importing especially for less developed countries (see Table 2 of Wagner 2012). In this study, by exploring the Turkish case we therefore expand the literature on self-selection into trade for emerging developing countries. In exploring the self-selection effects at work we control for the importing status of exporting firms and vice versa which is commonly neglected in the debate. This strategy enables us compare the strength of self-selection effects associated with importing and exporting activities<sup>7</sup>. Secondly while exploring the role of self-selection effects, and unlike previous papers, we take variable costs into account. Accordingly, we assess the impact of including variable costs (those associated with tariffs) on the size of the estimated sunk costs, and show that including these costs does impact on the results. Thirdly and building on the literature suggesting a link between productivity and product complexity, we investigate the differentials between sunk costs for importing/exporting of capital, intermediate and consumption goods. Finally, we assess and find a differential role for the product and country intensive margins on the productivity of importers and exporters.

Overall, in line with existing evidence we show that firms that engage in both sides of the trading activities perform better than the ones involved only in one side of trade, whereas all types of internationalized firms outperform the noninternationalized firms. Our findings also suggest that obtaining more varieties of imported intermediates (either in terms of numbers of products or countries) has a bigger impact on firm performance than exporting to more countries or exporting more products.

The distinction between exporters and importers provides further evidence on the remarkable heterogeneity across firms, where only-importers (importers) perform better than only-exporters (exporters). Observing a more persistent behaviour for importers with respect to exporters, our data suggest higher sunk costs for importing activity than for exporting. Indeed, we detect a self-selection effect

where the EU and EFTA lose grounds towards new markets in the Middle East and North Africa (MENA) as well as in Europe and Central Asia.

<sup>&</sup>lt;sup>7</sup>Such a comparison is crucial for firms operating in Turkish manufacturing industry for whom the most significant characteristic is its dependence on imported intermediary goods. For instance, in 2010 the imported component of Turkish manufacturing industry was 40 percent. Furthermore, in 2010 the growth of imports for manufacturing has surpassed the growth rate of manufacturing itself, implying that the dependency of the manufacturing industry on imports has increased. Sectors that grow above the average industry growth of Turkey typically have larger share of import component.

for both importing and exporting firms with a stronger effect for importers. In contrast with much of the literature which has failed to control for importing status of exporting firms and vice versa, when we take trading status of firms into account, we find that the self-selection effect is still present but greatly reduced with a smaller reduction for importers compared to exporters.

Next, employing a dynamic approach we account for sunk costs by means of past-trade experience and show that nature of sunk costs varies between importing and exporting activities, and that Turkish manufacturing importers face higher sunk costs in. Moreover, in contrast with the previous empirical literature which fails to control for variable costs of trade, our results signals that such costs may indeed constitute an important part of the story. Once we take tariffs which is an important component of variable costs into account, we find that the sunk costs for importing and exporting declines, with a smaller reduction for importers compared to exporters. We further show that the sunk costs are higher for capital goods, than intermediate and consumption goods for both of trading activities with higher sunk costs for importers in terms of each category.

The remainder of this paper is organized as follows. Section two briefly reviews the existing literature while section three introduces the data used in the empirical investigation. Section four gives some descriptive evidence on trading status dynamics, intra and inter-sectoral concentration and country and product extensive margins of exports and imports. Section five presents the empirical results. Section six concludes.

# 2 The Literature

The international trade literature has witnessed a dramatic change over the past eighteen years after Bernard and Jensen (1995) and Aw and Hwang (1995). These studies attracted the focus to the firm level analyses from country and industrylevel studies. The initial microeconomic empirical literature examining international trade at firm level reveals that exporting firms perform better than nonexporters. More recently, further evidence of firm heterogeneity related to firms' importing activities is put forward (Halpern et al. 2005; Bernard et al, 2007; Kasahara and Lapham, 2008).

The empirical findings of the literature on firm performance and trading activities reached their solid theoretical basis with Bernard et al.'s (2003) and Melitz's (2003) general equilibrium models. These studies explain the mechanism of most productive firms' self-selection into export markets. According to their models, there is substantial heterogeneity of firms within narrow industry borders in terms of productivity, size and other firm characteristics. Melitz (2003) builds his monopolistic competition model onto the assumption that there exist additional costs

for the firms selling in international markets. Therefore, only the firms surpassing some threshold level of productivity can make positive profits in international markets. These costs, which are defined as sunk costs, are related to transportation and establishing new distribution channels. They constitute entry barriers and hence only the most productive firms self select into exporting. On the other hand, Bernard et al. (2003) advocates that self-selection into exporting occurs via variable trade costs. Moreover, market size and these variable costs can create self selection of productive firms into foreign markets regardless of the existence of sunk costs. There is a vast empirical evidence supporting self selection hypothesis (Roberts and Tybout, 1997; Bernard and Jensen, 1999; Aw et al., 2000; Bernard and Wagner, 1997; Isgut, 2001, Delgado et al., 2002). Another observation of the regarding literature is that exporters tend to pay higher wages and benefits. Some scholars argue that this wage premia is a result of self selection into exporting. That is, if more productive firms self select into foreign markets it is natural to expect these future exporters to pay ex ante higher wages. For instance, Schank et al. (2010) show that wage premia exist for exporters some years before entry to the export market in Germany<sup>8</sup>.

Besides the self selection mechanism indicating ex-ante productivity differences, some scholars explain the performance premia of exporting firms by learning by exporting hypothesis. This hypothesis originates from Arrow's (1962) learning by doing model. It suggests that exporting improve firm performance both via interaction with the foreign clients and being exposed to intensive competition in the international markets. This process creates positive learning effects pushing firms to the efficiency frontier with respect to the non-internationalized firms. Several studies find support for post entry mechanism (see among others Castellani, 2002; Baldwin and Gu, 2004; Girma et al., 2004; Van Biesebroeck, 2005; Isgut and Fernandes, 2007; Lileeva and Trefler, 2007; Serti and Tomasi, 2009; Maggioni, 2012). However, this evidence is numerically less, controversial and conditional on special circumstances compared to the self selection mechanism<sup>9,10</sup>. Regarding wage premia, from the point of learning by exporting; it is hypothesized that

<sup>&</sup>lt;sup>8</sup>See also Serti et al. (2010) for Italy; Tsou et al. (2006) for Taiwan; Kandilov (2009) for Chile; and Amiti and Davis (2008) for Indonesia.

<sup>&</sup>lt;sup>9</sup>For instance, while Clerides (1998) find strong support for the self selection hypothesis, he finds no evidence for the learning by exporting in Colombia, Mexico and Morocco. In adition, in their emprical study Aw et al (2000) reveals that leraning by exporting mechanism is evident in Taiwan but not in South Korea. In a different manner, in their empirical investigation of Spanish manufacturing firms Delgado et al. (2002) finds evidence for learning by exporting mechanism only for a supsample of young firms. In a cross country comparison, The International Study Group on Exports and Productivity (2008) finds hardly any evidence on learning by exporting while suggesting self selection for 14 European countries.

<sup>&</sup>lt;sup>10</sup>For a detailed survey of the learning-by-exporting literature see Silva et al. (2010) and see Martins and Yang (2009) for a detailed analysis of 33 empirical studies.

exporting improves firm productivity leading to higher wages (Bernard and Jensen, 1999; Bernard and Wagner, 1997; Baldwin and Gu, 2004).

While there is a vast literature on the issue of firm heterogeneity from the exporting side, there are limited numbers of studies dealing with importing activities<sup>11</sup>. Although the literature is not yet well established, the literature on import premia also focuses on the self-selection mechanism of importing (positive effect of productivity on importing) and post-entry mechanisms (positive effect of importing on productivity). Self selection hypothesis builds on the observation that due to fixed costs of importing, only the firms above some productivity threshold could import where firms with high productivity levels offshore their production while low productivity firms limit themselves to sourcing from domestic markets. Similar to the self selection mechanism in the exporting case although the nature of these costs are different; they are also referred as sunk costs (e.g. search costs for foreign suppliers, inspection of goods, negotiation, contract formulation, learning and acquisition of customs procedures). Particularly, importers face greater informational asymmetries associated with imperfect monitoring of the purchased goods quality and cost of transferring the technology embedded into it (Altomonte and Békés, 2009). There is limited evidence on the self-selection into importing. Vogel and Wagner (2010) for Germany, Eriksson et al. (2009) and, Smeets and Warzynski (2010) for Denmark, Altomonte and Békés (2010) for Hungary find supporting evidence for the self selection mechanism of importing.

Regarding the post entry mechanisms of importing theoretical models (Grossman and Helpman, 1991; Eaton and Kortum, 2001; Acharya and Keller, 2007) emphasize a strong causality effect from importing intermediate and capital goods to firm performance via international knowledge spillovers, variety effect (higher productivity due to the access to more variety of inputs) and a quality effect (utilizing better quality inputs than domestic ones). Indeed, while there exists a large number of empirical studies for importance of knowledge spillovers through imports at the country and industry level<sup>12</sup>, there remains limited number of studies at firm level. Kasahara and Lapham (2008) for Chile, Bas and Strauss-Kahn (2010) for France, Forlani (2010) for Ireland and, Paul and Yasar (2009) and Dalgıç et al. (2014) for Turkey provides empirical evidence for the post entry mechanisms of importing.

Firm heterogeneity entered the empirical literature with evidence on two way trading activities as well. In particular, there is a number of empirical studies combining firms' importing and exporting activities by classifying them as exporters only, importers only or two-way traders. The general finding of this recently devel-

<sup>&</sup>lt;sup>11</sup>Early empirical literature focused exculusively on firms' exporting activities since exporting information of firms were reported more properly and comprehensively in censuses of production (Foster-McGregor, 2014).

 $<sup>^{12}</sup>$ See the seminal works of Coe and Helpman (1995) and Coe et al. (1997).

oped empirical literature is that two-way traders are more productive than firms that either only import, only export, or do not trade at all (Muuls and Pisu, 2009; Andersson et al., 2008; Kasahara and Lapham, 2008; Vogel and Wagner, 2010; Smeets and Warzynski, 2010; Serti and Tomasi, 2009; Altomonte and Békés, 2009; Silva et al., 2012; Castellini et al., 2010, Foster-McGregor et al., 2014). Furthermore, there is a hierarchy in firm performance generally from two-way traders to importers and then to exporters, while the non-internationalized firms perform the worst.

Firm heterogeneity in international trade literature also tackles the issue at the dimension of concentration as well as firms' geographical and product diversification i.e. country and product extensive margins (see Mayer and Ottoviano, 2007). This view points out that trade is concentrated in a few firms within an industry where concentration of trade is followed by unequal productivity distribution of firms. For example, Bernard et al. (2007) reports that in US a high percentage of export volume is performed by a small number of firms which are very diversified in terms of products and destination countries. Muuls and Pisu (2009) also present evidence that exports of Belgium rely largely on a small number of firms. Empirically a similar pattern is proven to be available for importing activities (e.g. Damijan et al., 2004; Andersson et al., 2008). Regarding diversification of trading activities, the proponents of self selection mechanism emphasize that firms which are more diversified in terms of country and product margins are more productive with respect to less diversified firms. This view is again supported by the idea of additional costs of selling goods in foreign markets. As the product and country range expands these additional costs are expanded and thus only a small number of firms with better performance can serve more variety of products to more countries (Muuls and Pisu, 2009; Wagner, 2007; Lawless, 2009).

# 3 Data

This paper relies on a recent dataset based on two different sources of data collected by Turkish State Institute of Statistics (TURKSTAT). The first one is The Annual Industry and Service Statistics and the second one is Annual Trade Statistics. The Annual Industry and Service Statistics is a census of firms with more than 19 employees while it is a representative survey for firms with less than 20 employees. For this study, we select the whole population of private Turkish manufacturing firms with 20 employees or more<sup>13</sup>. It includes information on entry, exit and

<sup>&</sup>lt;sup>13</sup>Firms with 20 and more than 20 employees account for a large share of Turkish manufacturing industry. For example, they constitute 87 percent of production in value and 75 percent of employment in 2009. It shows a similar pattern in the previous and following years. Moreover in the presence of sunk costs since trade activity is mostly performed by large firms our selection

missing values of some variables of the firms as well. In Annual Industry and Service Statistics dataset, firms are classified according to their main activity, as identified by Eurostat's NACE Rev.1.1 standard codes for sectoral classification<sup>14</sup>.

The database provides detailed information on a number of structural variables which are mainly seen on a firm's balance sheet such as revenues, value added, labour cost, intermediate inputs cost, tangible and intangible investment costs<sup>15</sup> together with information on industry and geographical location, foreign ownership and the number of employees. We calculate capital stock series of firms applying the perpetual inventory methodology and using the data on investment cost series for machinery and equipment, building and structure, transportation equipment and computer and programming<sup>16</sup>.

The second source of data we utilize is firm level foreign trade flows, which are sourced from customs declarations. The import and export flows are collected for the whole universe of the imports and exports at 12-digit GTIP classification the first 8 digits of whom correspond to CN classification whereas the last 4 digits are national. The information on the origin/destination countries of trade flows is also available in the dataset.

In order to conduct our analyses on firm level heterogeneity and international trade status of firms we merge annual industry and service statistics with annual foreign trade statistics. Our unbalanced panel covers longitudinal data of 38223 firms over the period 2003-2010<sup>17</sup>. Our sample is mainly constituted by small (62 percent) and small-medium firms (17 percent) whereas the rest of them are medium-large (21 percent)<sup>18</sup>.

<sup>16</sup>Initial capital stock is calculated by assuming that firms are at their balanced growth path and it is obtained by dividing the initial investment flow to the sum of depreciation rate and growth rate of output. For firms that reportzero investment in their initial year, it is assumed that they cannot be producing without capital. Therefore, initial capital stock is calculated at the year they reportpositive investment and this amount is iterated back to the beginning year by dividing capital stock each year.

<sup>17</sup>The original sample size in the merged dataset was slightly larger but we applied a cleaning procedure which is largely inspired by Hall and Mairesse (1995). We threw out the abnormal observations (zero / negative) for the main variables such as output, intermediate inputs, labor cost etc. Then, we excluded observations where main variables and ratios (e.g. employee, value added per employee, capital per employee) display extraordinary jumps and drops over one year. Finally, we excluded firms in NACE sectors 16 (Manufacture of tobacco products), 23 (Manufacture of coke, refined petroleum products and nuclear fuel), 30 (Manufacture of office, accounting and computing machinery), 37 (Recycling) since they include small number of firms.

<sup>18</sup>Firms with the number of employees 20-50 are defined as micro, 51-100 are defined as small,

does not create biased results.

<sup>&</sup>lt;sup>14</sup>The economic activities that are included in the survey are the ones in the NACE sections from C to K, and from M to O.

<sup>&</sup>lt;sup>15</sup>All nominal values are deflated using 4-digit NACE price indices with the base year 2003. For capital goods we use an aggregate investment deflator provided by the Ministry of Development. Wages are deflated by consumer price index.

Table 1 presents the number of firms and total number of employees in each year. On average we have 17000 firms over the analysis period. There is a big growth in the number of firms over 2003-2010. Accordingly, we observe that between the starting (2003) and the end period (2010) the entire sample of manufacturing firms has increased by 42 percent. The total number of employees hired by these firms was over 1232802 at the beginning of the period and reached 1957774 towards the end of the period. It is not surprising to observe a significant slump in the sample size in 2009 since Turkish economy was seriously hit by the global crisis in 2008.

### Insert Table 1 here.

We utilize different indicators to assess the firm heterogeneity according to their trade status. We use two different measures for firm-level productivity. One is total factor productivity (TFP) calculated using the Levinsohn and Petrin's (2003) approach<sup>19</sup>. The other is the standard labour productivity (LP), defined as value added (gross output net of intermediate inputs) per employee. To measure the scale of operation or size we utilize total manufacturing sales (Sales) and number of employees (Employee). We define capital intensity (Capint) of the firm as the ratio of the capital stock to the number of employees. To proxy skill intensity, we use wage per employee (Wage\_L).

# 4 Preliminary Evidence

In order to explore the linkages between firm characteristics and the internationalization status of firms we first classify the firms according to their trading status. We define the firms serving in the domestic market only as 'non-traders'; the firms engaged in exporting activities (including those that only export and those that not only export but also import) as 'exporters'; firms engaged in importing activities (including those that only import and those that combine their imports with exporting activities) as 'importers'; firms that do not export or import separately but are simultaneously involved in exporting and importing activities as 'two-way traders'. We define 'only-exporters' and 'only-importers' as well.

In Table 2, we provide descriptive evidence on our manufacturing industry panel, differentiating firms according to their participation in foreign markets. Over the period 2003-2010, on average 64 percent of all firms are internationalized. Two-way traders representing 39 percent of the whole sample constitute the

<sup>101-250</sup> are defined as medium and 250+ are defined as large.

<sup>&</sup>lt;sup>19</sup>Levinsohn and Petrin's (2003) methodology is a semi-parametric approach where TFP is measured as the residual of labour and capital under Cobb-Douglas technology, employing the firms' usage of intermediate inputs as a proxy variable for unobserved productivity shocks.

largest share of internationalized firms, while firms that engage in only exporting or only importing activities are a minority (only exporters, 11 percent and only importers, 13 percent). Exporting firms constitute 50 percent of the panel whereas importing firms' share is slightly higher with 52 percent on average. Different sample definitions in the related empirical literature make cross-country comparisons difficult. Nevertheless, Silva et al.(2013) working with Portugese data reveals a similar structure. In contrast other studies/contexts provide evidence on the relatively small or high share of exporting firms. For instance, Bernard et al. (2007) reports the share of exporters in the US manufacturing industry as 18 percent while Andersson et al. (2008) reports the share of exporters among whole Swedish firms as 83 percent. Bas et al. (2010) and Castellani et al. (2010) find somewhat different shares of exporters for France and Italy compared to our work. With an unrestricted sample, for Belgium, Muuls and Pisu (2009) reports the share of exporters as 41 percent, whereas with a restricted sample of firms with more than 20 employees ISGEP (2008) show that this share doubles to 84 percent.

Throughout our period of analysis the distribution of firms according to trading status stays fairly constant. For instance, the share of only-exporters stays in a range between 8.5-12 percent while share of importers stays in a range between 12.1-14  $percent^{20}$ . Column three of Table 2 shows that two-way traders are the most likely group to preserve their status. Yet, there is quite a lot of churning in terms of entry and exits. The share of entrants in 2010 with respect to 2003 is 94.5  $percent^{21}$ . The share of entrants is highest in only-exporters category, where the smallest share of entry has been realized by only-importers. Firms that were active in 2003 but not in 2010 (i.e. exiting firms/deaths) are evident in all categories with a share of 51.8 percent in total. Over the analysis period, the largest share of such exits has been realized by non-internationalized firms, which is consistent with the theoretical and empirical view that non-traders are at the lowest part of the productivity distribution (see Bernard et al., 2003; Melitz, 2003). Consistently, the smallest share of deaths is realized by the firms engaging in both sides of the trading activities which are shown to be at the highest part of the productivity distribution. Additionally, the rate of exits is higher for only-exporting firms compared to the only-importers (49.4 percent for only-exporters vs. 43.6 for the latter). This

<sup>&</sup>lt;sup>20</sup>However, compared with previous studies with former data on Turkey the shares in this study are higher. For instance, our data also suggest that the share of exporters stays in a range between 46-54 percent while share of importers stays in a range between 49-54 percent. Maggioni (2012) finds this share of exporters to be in the range 25-31 percent, whereas the share of importers is found to be ranged between 26-33 percent over 1990-2001. This indicates the increasing internationalisation of Turkish firms

<sup>&</sup>lt;sup>21</sup>For instance, there are 14788 firms in 2003 while 13968 firms entered to the sample throughout 2003-2010.

evidence might be attributable to higher productivity thresholds for only-importers relative to those of only-exporters, and for which we provide evidence later in this paper. Finally, Table 2 also presents that internationalized firms create a large share of employment in Turkish manufacturing industry.

### Insert Table 2 here.

Table 3 presents the dynamics of continuing internationalized firms' switching between trading categories. Movements of firms between trading categories shows significant variation. We observe that it is easier for both only-importers and only-exporters to alternate between the two kinds of trading activities. Moreover, starting to trade as a two-way trader is a rare event for a non-trader whereas stopping to trade for a two-way trader is the least likely outcome.

### Insert Table 3 here.

It is also interesting to see the trading status dynamics over the global crisis period  $(2007-2009)^{22}$ . In 2009, the number of firms shrank sharply with the 2008 global crisis. Overall, over the period 2007-2009 the number of firms fell by 17 percent. When considering performance differentials in 2007, we find that the exiting firms are found to be smaller and less productive compared to the survivors. The main driver of the fall in the number of firms are the non-internationalized firms while the impact on two-way traders is minor<sup>23</sup>. Note also, that in our sample foreign affiliation is a significant property of internationalized firms, hence trading activities may linked to the intra-firm international fragmentation of production. Among the foreign affiliated firms approximately 85 percent are exporters.

The evidence highlights that trade is more concentrated than employment or sales (See also Castellini et al. (2010) for Italy; Muuls and Pisu (2009) for Belgium; Bernard et al. (2007) for US; Silva et al. (2013) for Portugal).<sup>24</sup> Compared with Silva et al. (2013) who utilize Theil indices for Portuguese manufacturing firms, trade is more concentrated in Turkey than. Unlike the Portuguese case but similarly to Belgium and Italy, trade concentration shows an increasing trend in over time. Investigating by sectors, while there is clear sectoral heterogeneity, trade is more concentrated than sales and employment, for every Turkish manufacturing sector. These findings could be attributable to inter-industry trade specialization

<sup>&</sup>lt;sup>22</sup> A variety of transition dynamics between alternative pairs of years is available upon request. <sup>23</sup>Our choice of the crisis period might seem arbitrary i.e. taking the last quarter of 2008 would be the best interval for defining the beginning of the crisis. However, since we do not have quarterly data, we defined the pre-crisis year to be 2007. Still, we also checked the firms' transitions from 2008-2009. The transition dynamics is similar for both choice of periods.

<sup>&</sup>lt;sup>24</sup>Although not presented here, Gini and Theil coefficients confirm this finding for both exports and imports over the period 2003-2010.

(where trade is concentrated in few sectors) and also intra-industry trade specialization (where within the sector a subset firms carry out most of the trade). When we decompose the Theil index, it is the intra-industry component of the Theil index that explains the largest proportion of the concentration of trade i.e. trade is typically concentrated in a handful of firms within an industry. Although inter-sectoral components of our inequality measures in terms of exports and imports are low, exports are found to be concentrated mainly in six sectors (food and beverages, textiles, apparel, machinery and equipment, motor vehicles, basic metals) representing on average 73 percent of total export volume. Similarly, imports are concentrated mainly in four sectors (textiles, chemicals, motor vehicles, basic metals) representing on average 62 percent of total import volume.

We also investigate the diversification of trade along product and country extensive margins. The product extensive margin refers to the number of products that a firm exports/imports whereas the country extensive margin refers to the number of countries a firm trades with (Eaton et al., 2004; Mayer and Ottaviano, 2007)<sup>25</sup>. On average Turkish manufacturing firms export 10 products<sup>26</sup> and to 7 countries<sup>27</sup>, whereas they import 17 products<sup>28</sup> and from 6 countries<sup>29</sup>. These figures are smaller relative to evidence for developed countries<sup>30</sup>. Turkish firms' diversification in terms of product and country extensive margins increase both for exports and imports between 2003 and 2009<sup>31</sup> with much less striking rates for imports. Moreover, the rate of diversification is much higher in terms of country extensive margins relative to product extensive margins.

There is also some evidence on a negative relationship between the extensive margins and number of firms. For instance Andersson et al. (2008) find that in Sweden as the number of countries and products that firms export (import) increases the number of exporting (importing) firms decreases. Muuls and Pisu (2009) for Belgium and Castellani et al. (2010) for Italy also find a similar relationship between the country and product extensive margins and number of firms. These empirical results are consistent with the theoretical view that exporters (importers) incur additional costs of engaging in foreign markets and thus only a small number of firms can exist in international markets. We confirm these stylized facts for Turkish firms as well. In Tables 4 and 5, we present the share of exporting firms

 $<sup>^{25}\</sup>mathrm{Product}$  correspons to a 6-digit HS category.

<sup>&</sup>lt;sup>26</sup>A maximum of 423 different types of export products (HS6) are available in our dataset.

<sup>&</sup>lt;sup>27</sup>A maximum of 110 different types of countries to export are available in our dataset.

<sup>&</sup>lt;sup>28</sup>A maximum of 759 different types of import products (HS6) are available in our dataset.

<sup>&</sup>lt;sup>29</sup>A maximum of 64 different types of countries to import from are available in our dataset.

<sup>&</sup>lt;sup>30</sup>Investigating Belgian manufacturing firms Muuls and Pisu (2009), report that on average 12 products are exported and 34 products are imported, while Bernard et al. (2005) report that on average exporters sell 8.9 products and importers buy 10 products in US.

<sup>&</sup>lt;sup>31</sup>We exlude 2010 from this growth calculations due to the crisis effect.

(importing firms respectively) along with country and product extensive margins in 2003 together with firms' share of trade volumes. We show that a small proportion of firms account for a high proportion of the value of trade and this can be seen both the product and country extensive margins. For instance, according to the upper panel of Table 4, in 2003 46 percent of all exporting firms serve in up to 5 countries and 5 products, whereas 2.5 percent of firms export more than 20 products to more than 20 countries. From the lower panel of Table 4 one can infer that this small share of firms performs approximately 41 percent of total export volume in Turkish manufacturing industry. Compared to the studies on other countries with extensive margins above 5, in general Turkish manufacturing firms seem to be more diversified. For instance, the percentage of firms that export more than 5 products to more than 5 countries is 20 percent for Hungary, 35 percent for France, 43 percent for Portugal, 50 percent for Sweden and 70 percent for Italy while in Turkey this percentage is 54 percent.

Insert Table 4 here.

Insert Table 5 here.

# 5 Empirical Analysis

## 5.1 Do internationalized firms perform better?

In this part of the paper we continue to focus on some stylized facts in the recent trade literature. Our simple descriptive statistics presented in Table 6, are fully in line with the big picture that emerges from the literature reviewed earlier. We show a clear ranking of firm types by performance from two-way traders to importers and then to exporters. In particular, we find that non-traders are less productive, are less capital intensive, smaller in terms of number of employees and sales and pay lower wages. Moreover, in terms of the criteria listed above twoway traders are the best performers. The discrimination between exporters and importers provides further evidence on the remarkable heterogeneity across firms. Only importers (importers) outperform only exporters (exporters).

### Insert Table 6 here.

Next, we provide some empirical evidence on traders' premia, i.e., we present performance differentials between non-traders and trading firms controlling for other factors that could impact on performance. For instance, it is well established that larger firms are on average more productive than smaller firms or foreign affiliated firms are on average more productive than firms that serve only to the domestic market. Furthermore, two-way traders are found to be larger and have a higher foreign affiliation share than non-traders. This raises the question whether the productivity differentials between non-traders and two-way traders arise simply because they are larger or have a higher foreign ownership share. To eliminate this bias, following Bernard and Jensen (1999) and several other studies, we explore the following relationship between firm level characteristics and international trading status with the OLS regressions presented below:

$$y_{it} = \alpha + \beta_1 D_{it}^{two-way} + \beta_2 D_{it}^{only-imp} + \beta_3 D_{it}^{only-exp} + \delta Controls + \varepsilon_{it}$$
(1)

Where the subscript i denotes individual firms and t indexes year. The dependent variable  $y_{it}$  measures the logarithm of either firms' sales, number of employees, labor productivity (LP), total factor productivity (TFP), capital intensity or wage per employee. Dummies for the trading status are denoted by  $D_{it}^{two-way}, D_{it}^{only-imp}$ and  $D_{it}^{only-exp}$ , respectively, dummy variables for a two-way trader, only importer and only exporters. We utilize a series of control variables denoted by the vector of controls including the logarithm of firm's employment, two-digit sector dummies, region<sup>32</sup> and year dummies. We also include foreign affiliation dummy, a dummy for the existence of foreign ownership as a control. The coefficients  $\beta_1$ ,  $\beta_2$ and  $\beta_3$  in front of the trading dummies in equation (1) reveal the average trading premia in terms of various performance indicators. The traders premia are computed from the estimated coefficients as  $100(exp(\beta) - 1)$ , shows the average percentage difference in performance indicators between a firm in one of the three respective groups of trading firms and the non-traders, controlling for the characteristics included in the vector of controls. Equation (1) is also estimated with firm specific time invariant fixed effects, in order to deal with unobserved aapects of firm heterogeneity

The results from the pooled OLS regressions and FE regressions are reported in Table 7. Supporting the descriptive evidence above, the trade premia in terms of productivity, size, capital intensity and wages are of considerable magnitude and statistically significant. Specifically, internationalized firms have higher productivity levels, have higher capital intensity, larger in terms of employment and sales and pay higher wages than non-trading firms even after controlling for size, region, sector and time effects. In terms of the underlined performance criteria the magnitude of the trade premia coefficient declines significantly in the FE specifications pointing to the role of unobserved heterogeneity and the importance of firm specific factors. For instance, in terms of TFP while two-way traders are

 $<sup>^{32}</sup>$ The region dummies identify 12 Turkish regions distrubuted according to the NUTS2 classification.

estimated to be 51 percent more productive than non-internationalized firms in the OLS specification, in the FE model this premia reduces to 14 percent.

In both the OLS and FE specifications, two-way traders have the highest premia for all performance indicators, followed by firms that only import, while firms that only export have the smallest estimated premia. Note that the hierarchy suggesting that two way traders perform best followed by only-importers, and then only-exporters and finally non-traders remains after the inclusion of time invariant fixed effects into equation 1. This performance ordering of firms is in line with general finding of the empirical literature using this workhorse model (Muuls and Pisu, 2009; Serti and Tomasi, 2009; Altomonte and Békés, 2009; Silva et al., 2012; Castellini et al., 2010) with a few exceptions of McCann (2009) and Vogel and Wagner  $(2010)^{33}$ . The fact that importers are more productive than exporters can be attributed two different but not mutually exclusive explanations. The first is to do with self-selection effects and fixed costs; and the second is to do with the possible impact of importing on productivity. Indeed, regarding the latter Dalgic et al. (2014) shows that importing has a greater impact on productivity compared to exporting in Turkish manufacturing industry where they construct treatment groups as firms that are involved only with import activities and only with export activities respectively $^{34}$ .

Regarding the former, the advocates of the self-selection hypothesis suggest that only more productive firms will be able to import due to the fixed costs of importing. That the evidence from both descriptive statistics and regressions reflect higher performance premia for only-importers (importers) than only-exporters (exporters), may also suggest a stronger self-selection mechanism associated with importing at work with respect to exporting. Put differently, the fixed costs associated with importing might be higher than those of exporting. We therefore analyze the existence of the self selection mechanism with a special focus on the question of whether a stronger self selection mechanism is at work for importing activities than exporting in Turkish manufacturing industry.

### Insert Table 7 here.

Note that, so far the analyses conducted provide correlations/associations between firm performance and international trade engagement as opposed to showing causality. The existing literature typically fails to employ dynamic specifications

<sup>&</sup>lt;sup>33</sup>McCann (2009) and Vogel and Wagner (2010) finds that only exporting firms outperform only importing firms.

<sup>&</sup>lt;sup>34</sup>They argue that this result is in line with the view that while importing intermediate and capital goods transfer foreign knowledge accumulation directly to the domestic production processes, learning by exporting is not an automatic process and exporting does not necessarily improve firms (see Albornoz and Ercolani 2007; Wagner 2012).

in to address possible issues of endogeneity.<sup>35</sup>. Therefore, in order to shed light on possible issues of endogeneity associated with the FE regressions, we test a dynamic specification, which also serves as a robustness check. Thus, we run a series of fixed effects regressions in which we incorporate the lagged dependent variable as an additional regressor. Including the lagged dependent variable may produce biased and inconsistent parameter estimates because of its correlation with the individual specific effects. In such cases, GMM estimators are generally used to account for this endogeneity source (Blundell and Bond 1998; Bond 2002). While a proper estimation procedure is to address this endogeneity problem via GMM methodology, in large samples as ours the standard results for the dynamic model indicate that the OLS levels estimator is biased upward, while the within-group estimator is biased downward (Bond 2002; Bernard and Jensen 2004). We report on the FE estimates with lagged dependent variables for equation 1 in Table 7. The results from the dynamic specifications are consistent with our previous finding indicating the positive correlation between firm performance and trade engagement as well as the clear pattern of performance ordering types of internationalization status. Further, the significant coefficient of the lagged dependent variables in these regressions confirms that a firm's performance history affects its current position.

# 5.2 Self-Selection and Sunk Costs: Exporting vs. Importing

When considering the relationship between firm performance and trading the issues of self-selection and post-entry mechanisms arise<sup>36</sup>. Tables 2 and 3 demonstrated that a substantial number of firms switch their internationalization status. This variation in our data signals the importance of identifying the self-selection mechanisms at work. In addition (i) in Table 2 we observe a more persistent behavior for importing firms with respect to exporters and, (ii) in Table 3 we observe that a higher percentage of importers switch to two-way trading than that of exporters switch to two-way trading. (i) and (ii) together suggest higher sunk costs for importing with respect to exporting in Turkey. In this part of the study, we therefore shed light on whether firms self select into trade and whether this effect is stronger for importing and we then shed light on the driving forces behind this.

We start with addressing the question whether being a trader is associated with firms' ex-ante superior performance. If more productive firms become traders then

<sup>&</sup>lt;sup>35</sup>Silva et al.(2013) is the only study that employs such a dynamic specification in this context.

 $<sup>^{36} \</sup>mathrm{In}$  terms of post-enty effects there exists evidence on higher efficiency gains from importing with respect to exporting in Turkish manufacturing industry (see Dalgiç et al. (2014) for further discussion) .

we should expect to find significant differences in productivity between future trade starters and future non-starters several years before entry. In order to do so, we define an only-export-starter as a firm which had never traded in the previous two years (t - 2 & t - 1) and starts to exporting only in year t. In this way, we can compare firms that did not trade internationally in years t - 2 & t - 1 and start to export in year t with firms that did not trade at all. Only-import-starters and twoway-starters are defined similarly. We thus have six cohorts and each corresponds to a year between 2005 and 2010. To explore the pre-entry differences in various performance indicators between trade starters and non-traders we estimate the following equation with the usual controls:

$$y_{it-\rho} = \beta_0 + \alpha_i + \beta_1 D_i^{Starter} + \delta Controls_{t-\rho} + \varepsilon_{it}, \text{ with } 1 \le \rho \le 2.$$

where  $D_i^{Starter}$  is a dummy variable taking value one if the firm is a starter and zero if the firm is always a non-trader. Results are reported in Table 8. The coefficients show the average percentage performance differential at t-2 between starters at t and firms with no international trade activity over the whole period. Overall, in line with the previous studies we find a self-selection effect for both importing and exporting firms<sup>37</sup>. Specifically, the results confirm that internationalized firms are ex-ante larger, more productive, more capital intensive and pay higher wages than non-traders. The performance premia is highest for two-way starters in terms of all criteria.

Note that, the pre-entry levels of the indicators are larger for only-import starters than those of only-export starters. For instance, two years before entering the import market, import starters are already approximately 32 percent more productive (in terms of TFP) than always non-traders<sup>38</sup> while export starters are 28 percent more productive than always non-traders. This suggests that importingonly firms exhibit ex-ante performance advantages with respect to those that export-only, in turn indicating a stronger self-selection for importing than exporting.

### Insert Table 8 here.

Failing to control for the importing status of exporting firms and vice versa might lead to overstating the role of self-selection in exporting and importing respectively. Thus, we further investigate the performance premia of future two-way traders compared to future only-exporters and future only-importers. In this way, we account for importers that start to export by comparing firms that imported

 $<sup>^{37}\</sup>mathrm{See}$  among others Kasahara and Lapham 2008; Altomonte and Bekes, 2009; Castellani et al., 2010.

<sup>&</sup>lt;sup>38</sup>The traders premia are computed from the estimated coefficients as  $100(exp(\beta) - 1)$ .

but not exported in years t - 2 and t - 1 and start to export in t with firms that always imported but not exported at all. Similarly, we investigate the performance premia of exporters that start to import. In the regressions presented in Table 8, the coefficients show the average percentage performance difference at t - 2 between only-exporters that start to import (only-importers that start to export) at t as well and only-exporters (only-importers) that do not start to import at all. We find that when taking into account the importing status of export starters, the performance premium of export starters is still present but it is greatly reduced. Similarly, the performance premium of import starters is still present but with a smaller reduction in the coefficient compared to the export starters.

Hence, taking into account the importing / exporting status of exporter / importers respectively serves to accentuate the higher productivity associated with importing in contrast to exporting firms. In addition, these findings indicate that initial pre-entry premia reported in Table 8 are likely to overstate the extent to which export and import starters had higher initial productivity levels. In particular, the inclusion of the importing decision lowers the pre-entry productivity premia from 28 to 7 percent for period t-2, while the performance premium of import starters declines less - from 32 to 21 with the inclusion decision. We therefore conclude that for Turkish manufacturing firms the self-selection effect is evident in both exporting and importing activities but is stronger with respect to importing. A limited number of studies control for the importing status of exporting firms or vice versa in investigating self-selection effect associated with entering into foreign markets. Following a similar analysis and using Hungarian data, Altomonte and Békés (2009) find that ex-ante productivity of importing is larger than that of exporting.

The evidence so far highlights that there is a stronger self selection effect at work for import starters compared to export starters. This might suggest sunk costs of importing are greater than that of exporting for Turkish manufacturing firms. Indeed, the recent literature on the self-selection mechanism provides insights for the possible heterogeneity of sunk costs across trading statuses of firms. While exporters assumed to face sunk costs linked to marketing and setup of foreign distribution channels importers do not face these typical costs. Importers are more likely to face greater informational asymmetries associated with the imperfect monitoring of the purchased goods quality and cost of using and transferring the technology embedded in their imports (see Altomonte and Békés, 2010).

Accordingly, we investigate the self-selection mechanism emphasizing the relative importance of the sunk costs and shed some light on the differentials between the sunk costs of importing and exporting. In order to do so, we estimate three dynamic models for firms that only-export, only-import and those involved in both activities. Following Roberts and Tybout (1997), Bernard and Jensen (2004) and Muûls and Pisu (2009), we account for sunk costs by means of past trade experience where the coefficient of the lagged dependent variable is interpreted as a measure of sunk  $costs^{39}$ .

Melitz and Ottaviano (2008) and Bernard et al. (2003) show that even in the absence of sunk costs most productive firms would still self select into exporting i.e., sunk costs may not be the sole determinant of self selecting into international trade. Accordingly, the lack of any performance controls for self selection process would lead to overstating the role of sunk costs. Thus we include lagged TFP, wage per employee and number of employees to account for past productivity performance, scale of operation and skill level respectively as well as controlling for endogeneity. We estimate the following random effects panel probit regression:

$$P(y_{it} = 1, x_{it}, y_{it-1}, u_i) = f(\alpha + \rho y_{it-1} + \beta' x_{it} + u_i)$$
(2)

where subscript *i* and index *t* denotes the individual firms and years, respectively. The binary variable  $y_{it}$  indicates whether the firm is a trader or not in one of three subsequent forms (exporting-only, importing-only or being a two way trader); *x* consists of our firm level performance controls including the mean of these controls as well as region, sector and year dummies;  $u_i$  captures the firm level unobservables where *f* denotes the cumulative normal distribution and where  $u_i$  can be expressed as <sup>40</sup>:

$$u_i = \beta_o + \beta_1 y_{i0} + \beta_2 \bar{x}_i + \epsilon_i \tag{3}$$

The results of the random effects dynamic probit model are presented in Panel A of Table 9. As standard in the literature, we confirm that the more productive and the larger the firms are, the more likely they self select into trade. Wage per employee is found to positively affect the probability of importing-only or being a two-way trader, yet it is surprisingly insignificant for only-exporters. We find that Turkish firms face sunk costs of engaging into international markets and the nature of these sunk costs varies between importing and exporting activities<sup>41</sup>. Specifically, the coefficient associated with lagged export status is lower than of

<sup>&</sup>lt;sup>39</sup>Kashara and Lapham (2008) built a theoretical expansion on Melitz (2003) and are the first to quantify the sunk costs of trading activities. They test their model for Chilean data and find higher sunk costs for exporting firms than importing.

 $<sup>^{40}</sup>$ In order to deal with the initial condition bias existing in dynamic limited dependent variable models and the possible correlation between the controls and unobserved heterogeneity we utilize Wooldridge's (2005) methodology which models firm specific effects  $u_i$  as a function of the initial condition and other explanatory variables. Accordingly, the model becomes a random effects probit model.

<sup>&</sup>lt;sup>41</sup>The initial trade status coefficients are high in magnitude and statistically significant correcting for the bias introduced by the 'initial condition' problem.

the importer coefficient suggesting that the sunk costs of importing-only are higher than the sunk costs of exporting-only for Turkish manufacturing firms.

### Insert Table 9 here

On the other hand, one possibility behind the self-selection mechanism might be linked to variable costs of trade. As in Melitz and Ottaviano (2008) and Bernard et al. (2003) higher variable costs of trading will mean only more productive firms will be able to enter into trade markets. That is they present different selection mechanisms based on variable trade cost instead of sunk costs of trading. In their model setting, market size and variable costs determine the toughness of competition and hence the strength of the self-selection effect. Data from the World Bank Doing Business Surveys suggests that there are indeed higher costs of importing. Exporting a standard container of goods requires larger number of documents, takes more time and costs higher for an importing firm than with respect to those of exporting<sup>42</sup>. Such data is not available neither at the product or bilateral levels. However, another key variable cost are the tariffs faced by the firms both with regard to importing and in export markets. In order to control for the variable costs of trading we re-run the dynamic probit regressions in Panel B of Table 9 including import and export tariffs<sup>43</sup> as additional controls.

The results in Panel B of Table 9 reinforces our previous finding that there is a stronger self-selection effect for importers than exporters. We see that when we control for tariffs, the coefficients representing the sunk costs for exporting and importing shrink to 0.921 and 0.959 from 0.878 and 0.949, respectively; and that the biggest narrowing takes place with regard to exporters. This suggests that the tariff-related variable cost elements is a more important component of the forces driving self-selection effect for exporters than with respect to importers. However, in addition, now the sunk costs of importing-only become relatively higher than previously in comparison to the sunk costs of exporting-only. Hence failing to consider the variable costs of trade may underestimate sunk cost differences.

Next, and given the previous finding that importing is associated with higher sunk costs we try and shed more light on the sunk costs that firms might face while selecting into trade markets. As mentioned before, Altomonte and Békés (2010) argue that importers face uncertainty in their trading relationships (e.g. with regard to the quality of the product). This uncertainty is likely to be higher the more complex is the good being traded; therefore fixed costs of trading are

<sup>&</sup>lt;sup>42</sup>The data suggests that exporting a standard container of goods requires 7 documents, takes 13.0 days and costs \$990.0. Importing the same container of goods requires 8 documents, takes 14.0 days and costs \$1063.0 in 2010. Over 2005-2012, the period in which the data is available, one can see that cost of importing in all dimensions is higher that that of exporting for Turkey.

<sup>&</sup>lt;sup>43</sup>Import and export tariffs at HS6 digit product category are collected from WITS-Trains database. We calculate firm level tariffs by weighting product-country level tariff information.

likely to be higher for more complex goods. They show that importers are more productive than exporters and associate this with higher import complexity. One way of looking at the complexity of goods is to classify them according to their final use. Therefore, we utilize United Nations' Classification by Broad Economic Categories (BEC) and define products traded in three broad categories as: consumption goods, intermediate goods and capital goods. Capital goods (e.g. machinery) are frequently more complex and may require after-sales service etc. with respect to other categories (Keller and Yeaple, 2008).

Descriptive evidence reveals that the share of capital goods imports in total imports is higher compared to capital exports in total exports for Turkish manufacturing industry thus Turkish imports seem to be more complex than exports. We distinguish between three types of firms: capital goods importers/exporters; intermediate goods importers/exporters and consumption goods importers/exporters. An only-importer (only-exporter) firm is defined to be capital goods importer (exporter) if the share of capital goods imports (exports) in its total value of imports (exports) is equal or greater than 0.5. We define other categories similarly.

Table 10 presents the random effects dynamic probit regressions run with these categories of firms in question. We show that the sunk costs are higher for capital goods, than intermediate than consumption goods for both importers and exporters. For instance, the coefficient of the lagged dependent variable associated with sunk costs of importing-only are 0.992, 0.961 and 0.874 for capital, intermediate and consumption goods importers respectively. While, the coefficients associated with the sunk costs of exporting-only are 0.933, 0.925 and 0.873 for capital, intermediate and consumption goods importers respectively.

As the sunk costs of capital goods are higher, thie lends support to the notion that this arises because of the higher complexity associated with such imports (as in Altomonte and Békés (2010)). Note that in each case these coefficients are higher for importers with respect to those for exporters. Once again these results reinforce our previous finding that sunk costs, to the extent that they drive selfselection, are more important in the case of importing than exporting in Turkey. The hierarchy of sunk costs from capital to consumption goods traded remains even after controlling for tariffs which are associated with variable costs. Another result from Table 10 is that, in terms of importing when tariffs are included as a control, the smallest decrease in the sunk cost coefficient occurs for capital goods, whereas for exporting the smallest decrease is with respect to intermediate goods. This suggests that variable costs are a smaller component of the costs leading to self selection mechanism for capital goods imports in comparison to intermediate or fianl goods; and that the strongest self selection effect in terms of sunk costs are with respect to capital goods imports.

### Insert Table 10 here

## 5.3 In search of diversification differentials: Exporting vs. Importing

The analysis so far has not addressed a key topic in the literature on firm heterogeneity in international trade, which assesses firms' the role of the diversification of firms in terms of geography and products. In this part of the paper, we therefore focus on those firms involved in both importing and exporting and explore the role of the country and product extensive margins in understanding diversification differentials between exporting and importing activities.

In Table 11, we compare the performance of two-way traders in terms of various firm characteristics. Here we group two-way traders according to their extensive margins. The first group consists of firms that trade less than 6 goods (countries), the second group consists of firms that trade 6-10 goods (countries), the third group consists of firms that trade 11-20 goods (countries) and the last group consists of firms that trade 11-20 goods (countries). We present the mean TFP and LP along extensive margins. Table 11 shows that the greater the number of either partners or products the higher is the level of productivity. For instance, the TFP of firms which export more than 20 products is on average 8 percent higher relative to the firms which export less than 6 products.

This differential is more significant on the import side than on the export side. Note Table 11 also shows that where the number of product or partners is low (e.g. 1-5), then the productivity is higher with regard to exporters than importers. As the number of either product or partners goes up, than the productivity associated with importers becomes higher than exporters. This could be linked to the hypothesis that more partners/products Turkey imports from, potentially the higher the fixed costs. This may be less of an issue on the export side, particularly as Turkey is largely exporting to EU markets with more open and easier access. Although the EU is also Turkey's major import partner, more distant countries such as China and United States have non-negligible shares in Turkey's imports<sup>44</sup>. In addition, on the import side, Turkey investment and intermediate goods comprise a high share of imports, and which are associated with higher sunk costs of trade relative to consumption goods<sup>45</sup>.

<sup>&</sup>lt;sup>44</sup>In 2013, top five export partners of Turkey are Germany, Iraq, UK, Italy and France whereas Russia, China, Germany, USA and Italy have the highest shares in Turkey's imports (Economic Outlook Report Ministry of Economy, 2013).

<sup>&</sup>lt;sup>45</sup>In 2013, the share of investment and intermediate goods in exports of Turkey is approximately 60 percent while that share is 88 percent in imports.

### Insert Table 11 here

However, in assessing the diversification of trade along country and product extensive margins one needs to control for other factors (e.g. sectoral or other firm characteristics) that could be associated with firm performance. Thus, we estimate the following specification controlling for the firm specific fixed effects:

$$y_{it} = \alpha_i + \beta_1 x_{it}^{NPE} + \beta_2 x_{it}^{NPI} + \beta_3 x_{it}^{NCE} + \beta_4 x_{it}^{NCI} + \delta Controls + \varepsilon_{it}$$
(4)

where the dependent variable  $y_{it}$  measures the logarithm of either firms' sales, number of employees, labor productivity, total factor productivity, capital intensity or wage per employee. The variable x denote the product and country extensive margins (NPE, NPI, NCE, NCI, respectively) in logarithms. The vector of controls includes the logarithm of firms' employment, two-digit sector dummies, region and year dummies. Each regression covers the sample of firms which are two-way traders throughout the analysis period. The coefficients  $\beta_1$ ,  $\beta_2$ ,  $\beta_3$  and  $\beta_4$  in front of the margin variables in equation 4, show the elasticity of our selected performance indicators with respect to extensive margins. These elasticities are interpreted as diversification premium of traders.

Table 12 indicates that the greater is the number of products or partners, the larger, more productive the firms are and the higher wages they pay. This effect is most significant with regard to the import side. Specifically, the diversification of imports along country and product extensive margins creates larger premia than exports<sup>46</sup>. For instance, changes in the number of products imported would have a higher impact on productivity than changes in the number of products exported. A 1 percent increase the number of products imported (NPI) is associated with approximately 5.5 percent increase in labor productivity while a 1 percent increase the number of products with only a 1 percent increase in labor productivity. In terms of capital intensity the diversification premia is positively significant for imports while it is found to be insignificant for exports.

These findings suggest that obtaining more varieties of imported intermediates (either in terms of numbers of products or countries) is associated with a bigger impact on productivity than exporting to more countries or exporting more products. The former impacts directly on efficiency in production; whereas the mechanisms driving the latter are presumably linked with economies of scope. In fact, the use of imported foreign capital and intermediate inputs which embody

<sup>&</sup>lt;sup>46</sup>In order to shed light on possible issues of endogeneity associated with FE regressions, we also test a dynamic specification as well as providing a robustness check. The results which are available upon request do not change substantially.

better technology is directly associated with technological upgrading and thus efficiency improvement (Damijan and Kostevc, 2010). In addition, the decision to invest in new technology can take place simultaneously with the decision of importing (Damijan et al., 2012). Both the existence of diversification premia and the more pronounced effect for imports results is also evident in Castellani et al. (2010) for Italian manufacturing firms and Silva et. al. (2013) for Portuguese manufacturing firms. They explain this differential as in order to enter new import markets, firms need to have the ability to value, assimilate and apply the new knowledge embodied in imports of high capital intensity.

Insert Table 12 here

# 6 Concluding Remarks

This paper uses a rich and recent dataset for Turkish manufacturing firms from 2003 to 2010 to provide the first comprehensive analysis of firm heterogeneity connecting firms' performances to international trade. More importantly, for the first time we investigate self selection into foreign markets systematically for Turkey and particularly focus on the differential among the nature of self-selection effect and the role of variable and sunk costs for importing and exporting.

Overall, in line with the big picture emerging from the existing literature we show that (i) a small proportion of firms account for a high proportion of the value of trade; (ii) firms that engage in both sides of the trading activities perform better than the ones involved only in one side of trade; (iii) all types of internationalized firms outperform the non-internationalized firms in Turkey. Descriptive analysis shows that although the distribution of firms according to their trade status stay fairly constant over the period in question, there is considerable churning in terms of entry and exit. The smallest share of exits is realized by firms engaging in both sides of the trading activities, with a higher rate of exits for only-exporting firms compared to the only-importers; suggesting higher productivity thresholds for only-importers relative to those of only-exporters.

Our preliminary regressions on trade premia also show a clear ranking of firm types by performance from two-way traders to importers-only and then to exporters-only. That the evidence from both descriptive statistics and regressions signal higher performance premia for only-importers (importers) than onlyexporters(exporters), which in turn may suggest a stronger self-selection mechanism associated with importing with respect to exporting.

Indeed, we confirm a self-selection effect for both importing and exporting firms with a stronger effect for importers in Turkey. While doing so we show that: (i) being a trader is associated with firms' ex-ante superior performance; (ii) the pre-entry levels of firm's performance indicators are larger for only-importers than those of only-exporters; (iii) the self-selection effect is still present but is somewhat reduced with a smaller reduction for importers compared to exporters after controlling for the importing status of exporting firms and vice versa; (iv) the nature of sunk costs varies between importing and exporting activities with importers facing higher sunk costs.

We show that the self-selection mechanism is associated with both variable and sunk costs. In particular, once we take the tariff related variable costs of trade into account, we find that the sunk costs for importing are even higher than for exporting. We further show that the sunk costs are highest for capital goods, than intermediate and consumption goods for both of trading activities, with higher sunk costs for importers in terms of each category.

# 7 Acknowlegments

We are grateful to the Turkish State Institute of Statistics (TURKSTAT) for providing access to firm-level data under a confidential agreement. In particular, we thank TURKSTAT staff Doğan Böncü, Nusret Kılıç, Nilgün Arıkan, Erdal Yıldırım, Kenan Orhan, Bülent Tungul, Akın Bodur and Sabit Cengiz Ceylan. An earlier draft was presented at the ETSG Conference 2013 in Birmingham.

# 8 Funding

We acknowledge the generous financial support of TÜBİTAK from the budget of the project entitled 'Firm Heterogeneity in Turkish Manufacturing Industry and International Trade' [project number 113K378].

# References

Acharya, R. and Keller, W. (2007), "Technology Transfer through Imports", NBER Working Paper No. 13086. URL: <u>https://ideas.repec.org/p/nbr/nberwo/13086.html</u>.

Amiti, M. and Davis, D.R. (2008), "Trade, Firms, and Wages: Theory and Evidence", NBER Working Paper No. 14106. URL: <u>https://ideas.repec.org/p/nbr/nberwo/14106.html</u>.

Andersson, M., S. Johansson, and H. Lööf (2008), "Productivity and International Trade: Firm Level Evidence from a Small Open Economy," *Review of World Economics*, 144(<u>4</u>), 774-801. URL: <u>https://ideas.repec.org/p/hhs/cesisp/0099.html</u>.

Arrow, K. (1962), "The Economic Implications of Learning by Doing," *Review of Economic Studies*, 29(3), 155-173. URL: <u>http://www.jstor.org/stable/2295952</u>.

Aw, B.Y., S. Chung and M.J. Roberts (2000), "Productivity and Turnover in the Export Market: Micro-level Evidence from the Republic of Korea and Taiwan (China)," *The World Bank Economic Review*, Vol.14, No.1, 65-90. URL: http://wber.oxfordjournals.org/content/14/1/65.abstract.

Aw, B.Y., and Hwang, A. (1995), "Productivity and the Export Market: A Firm-Level Analysis," *Journal of Development Economics*, 47, 313-332. URL: https://ideas.repec.org/a/eee/deveco/v47y1995i2p313-332.html.

Baldwin, R. and Gu, W. (2004), "Trade Liberalization: Export-market Participation, Productivity Growth, and Innovation", *Oxford Review of Economic Policy*, 20(3), 372-392. URL: <u>https://ideas.repec.org/p/stc/stcp5e/2004027e.html</u>.

Bas, M., and V. Strauss-Khan (2010), "Does importing more inputs raise exports? Firm level evidence from France," MPRA Paper No. 27315, December. URL: <u>https://ideas.repec.org/p/cii/cepidt/2011-15.html</u>.

Bernard, AB., Eaton, J., Jensen JB. and Kortum, SS. (2003), "Plants and Productivity in International Trade", *American Economic Review*, 93:1268–90. URL: <u>https://ideas.repec.org/p/nbr/nberwo/7688.html</u>.

Bernard, A. and B. Jensen (1999), "Exceptional Exporter Performance: Cause, Effect, or Both?" *Journal of International Economics*, 47, 1, 1–25. URL: <u>https://ideas.repec.org/a/eee/inecon/v47y1999i1p1-25.html</u>.

Bernard AB., and Jensen J.B. (2004) "Why Some Firms Export", *Review of Economics and Statistics*, 86:561–69. URL: http://www.jstor.org/stable/3211647?seq=1#page\_scan\_tab\_contents.

Bernard, A., B. Jensen, S. Redding and P. Schott (2007), "Firms in International Trade", *Journal of Economic Perspectives*, 21(3), 105–30. URL: https://ideas.repec.org/a/aea/jecper/v21y2007i3p105-130.html. Bernard, A., Jensen, J., Redding, S., and P. Schott (2009). "The Margins of US Trade (Long Version)", NBER Working Paper, 14662. URL: <u>http://www.nber.org/papers/w14662</u>.

Castellani D. (2002), "Export Behavior and Productivity Growth: Evidence from Italian Manufacturing Firms", *Review of World Economics*, 138(4), 605-628. URL: <u>https://ideas.repec.org/a/spr/weltar/v138y2002i4p605-628.html</u>

Castellani, D., F. Serti, and C. Tomasi (2010), "Firms in International Trade: Importers' and Exporters' Heterogeneity in Italian Manufacturing Industry", *The World Economy*, 33(3), 424–457. URL: <u>https://ideas.repec.org/a/bla/worlde/v33y2010i3p424-457.html</u>.

Clerides, S., S. Lach, and J. Tybout (1998), "Is Learning by Exporting Important? Microdynamic Evidence from Colombia Mexico and Morocco", *Quarterly Journal of Economics*, 113(3), 903-948. URL: http://www.jstor.org/stable/2586877?seq=1#page\_scan\_tab\_contents.

Delgado, M. A., J. C. Fariñas, and S. Ruano (2002), "Firm Productivity and Export Markets: a Non-parametric Approach," *Journal of International Economics*, 57, 397–422. URL: <u>https://ideas.repec.org/a/eee/inecon/v57y2002i2p397-422.html</u>.

Eaton, J. and Kortum, S. (2001), "Technology, Trade, and Growth: A Unified Framework," *European Economic Review*, 45(4-6), 742-755. URL: https://ideas.repec.org/a/eee/eecrev/v45y2001i4-6p742-755.html.

Eriksson, T., Smeets, V. and Warzynski, F. (2009), "Small Open Economy Firms in International Trade: Evidence from Danish Transaction-Level Data", Aarhus School of Business Department of Economics Working Paper 09-7, March.

Forlani, E. (2010), "Irish Firms' Productivity and Imported Inputs", CORE Discussion Paper 2010/15, April. URL: <u>https://ideas.repec.org/p/cor/louvco/2010015.html</u>.

Foster-McGregor, N., Isaksson, A. and Kaulich, F. (2014), "Importing, Exporting and Performance in Sub-Saharan African Manufacturing Firms," *Review of World Economics*, 150, 309-336. URL: <u>http://link.springer.com/article/10.1007%2Fs10290-013-0177-y</u>.

Girma, S., Greenaway, D., and Kneller, R. (2004), "Does Exporting Increase Productivity? A Microeconometric Analysis of Matched Firms," *Review of International Economics*, 12, 855-866. URL: <u>https://ideas.repec.org/a/bla/reviec/v12y2004i5p855-866.html</u>.

Greenaway, D. and R. Kneller (2007), "Firm Heterogeneity, Exporting and Foreign Direct Investment", *Economic Journal*, 117, 134-161. URL: http://onlinelibrary.wiley.com/doi/10.1111/j.1468-0297.2007.02018.x/abstract.

Grossman, G.M. and Helpman E. (1991), "Trade, Knowledge Spillovers, and Growth", *European Economic Review*, 35(3), 517-526. URL: <u>http://www.sciencedirect.com/science/article/pii/001429219190153A</u>.

Halpern, L., Koren, M., and Szeidl, A. (2005), "Imports and Productivity," CEPR Discussion Papers, 5139. URL:

http://www.cepr.org/active/publications/discussion\_papers/dp.php?dpno=5139.

International Study Group on Export and Productivity (ISGEP) (2008), "Exports and Productivity. Comparable Evidence for 14 Countries", *Review of World Economics*, 144, 4, 596–635.

Isgut, A., (2001), "What's Different about Exporters? Evidence from Colombian Manufacturing", *Journal of Development Studies*, 37(5), 57-82. URL: https://ideas.repec.org/a/taf/jdevst/v37y2001i5p57-82.html

Isgut, A. and A. Fernandes (2007), "Learning-by-exporting Effects: Are They For Real?" MPRA Working Paper 3121, Munich Personal RePEc Archive, University Library of Munich, Germany. URL: <u>https://ideas.repec.org/p/pra/mprapa/3121.html</u>

Kandilov, I.T. (2009), "Do Exporters Pay Higher Wages? Plant-level Evidence from an Export Refund Policy in Chile", *World Bank Econonomic Review*, 23(2), 269-294. URL: <u>http://elibrary.worldbank.org/doi/abs/10.1093/wber/lhp004</u>.

Kasahara, H. and B. Lapham (2008), "Productivity and the Decision to Import and Export: Theory and Evidence", CESifo Working Paper 2240, CESifo Group Munich. URL: <u>https://ideas.repec.org/p/ces/ceswps/\_2240.html</u>.

Lawless, M. (2009), "Firm Export Dynamics and the Geography of Trade," *Journal of International Economics*, 77 (2), 245-254. URL: https://ideas.repec.org/a/eee/inecon/v77y2009i2p245-254.html.

Levinsohn, J. and A. Petrin (2003), "Estimating Production Functions using Inputs to Control for Unobservables," *Review of Economic Studies*, 70, 317-42. URL: <u>http://www.jstor.org/stable/3648636?seq=1#page\_scan\_tab\_contents</u>.

Lileeva, A. and D. Trefler (2007), "Improved Access to Foreign Markets Raises Plant-level Productivity", Working Paper 13297, NBER. URL: <u>www.nber.org/papers/w13297</u>.

Maggioni D., (2010). "Learning by Exporting in Turkey: An Investigation for Existence and Channels", MPRA Paper No. 37157. URL: https://ideas.repec.org/a/bpj/glecon/v12y2012i2n5.html.

Martins, P.S., and Yang, Y. (2009), "The impact of exporting on firm productivity: a metaanalysis of the learning-by-exporting hypothesis", *Review of World Economics*, 145 (3), 431-445. URL: https://ideas.repec.org/a/spr/weltar/v145y2009i3p431-445.html.

Mayer, M. and G. Ottaviano (2007), "The Happy Few: The Internationalisation of European Firms. New Facts Based on Firm-level Evidence", Working Paper Volume 3, Bruegel Blueprint Series. URL: <u>http://dems.unimib.it/corsi/678/altro/the\_happy\_few.pdf</u>.

Melitz, M.J. (2003), "The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity", *Econometrica*, 71(6), 1695–1725. URL: http://www.jstor.org/stable/1555536?seq=1#page\_scan\_tab\_contents. Muuls, M., and Pisu, M. (2009), "Imports and Exports at the Level of the Firm: Evidence from Belgium", *The World Economy*, 32(5), 692-734. URL: <u>https://www.ecb.europa.eu/home/pdf/research/compnet/Muuls\_Pisu\_2009.pdf??94bb954a2c2</u>246ce182f83cc82d239ca.

Paul, C. J. M., and M. Yasar (2009), "Outsourcing, Productivity, and Input Composition at the Plant Level," *Canadian Journal of Economics*, 42(2), 422–439. URL: <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1540-5982.2009.01514.x/abstract</u>.

Redding, S.J. (2010), "Theories of Heterogeneous Firms and Trade", National Bureau of Economic Research NBER Working Paper 16562, December. URL: <u>www.nber.org/papers/w16562</u>.

Roberts, M.J. and Tybout, J.R. (1997), "The Decision to Export in Colombia: An Empirical Model of Entry with Sunk Costs," *American Economic Review*, 87, 545-564. URL:

Schank, T., Schnabel, C., and Wagner, J. (2010), "Higher Wages in Exporting Firms: Self-Selection, Export Effect, or Both? First Evidence from Linked Employer-Employee Data," *Review of World Economics*, 146(2), 303-322. URL: <u>www.jstor.org/stable/2951363</u>.

Serti, F. and C. Tomasi (2008), "Self-selection and Post-entry Effects of Exports. Evidence from Italian Manufacturing Firms," *Review of World Economics*, 144(4), 660-94. URL: <u>http://link.springer.com/article/10.1007%2Fs10290-008-0165-9</u>.

Serti, F., Tomasi, C., and Zanfei, A. (2010), "Who Trades with Whom? Exploring the Links between Firms' International Activities, Skills, and Wages," *Review of International Economics*, 18(5), 951–971. URL: <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1467-9396.2010.00919.x/abstract</u>.

Silva, A., Afonso, O. and Africano, A.P. (2013), "Economic Performance and International Trade Engagement: The Case of Portuguese Manufacturing Firms," *International Economics and Economic Policy*, 10, 521–547. URL: http://link.springer.com/article/10.1007%2Fs10368-012-0221-8

Smeets, V. and F. Warzynski (2010), "Learning by Exporting, Importing or Both? Estimating Productivity with Multi-product Firms, Pricing Heterogeneity and the Role of International Trade", Working Papers 10-13, mimeo, University of Aarhus. URL: <u>https://ideas.repec.org/p/hhs/aareco/2010\_013.html</u>

Tsou, M., Liu, J. and Huang, C. (2006), "Export Activity, Firm Size and Wage Structure: Evidence from Taiwanese Manufacturing Firms", *Asian Economic Journal*, 20(4), 333-354. URL: <u>http://onlinelibrary.wiley.com/doi/10.1111/j.1467-8381.2006.00242.x/abstract</u>

Van Biesebrock, J. (2005), "Exporting Raises Productivity in Sub-Saharan African Manufacturing Plants," *Journal of International Economics*, 62(2), 373-391. URL: <u>https://ideas.repec.org/a/eee/inecon/v67y2005i2p373-391.html</u>

Wagner, J. (2007), "Exports and Productivity: A Survey of the Evidence from Firm Level Data", *The World Economy*, 30(1), 60-82. URL: http://onlinelibrary.wiley.com/doi/10.1111/j.1467-9701.2007.00872.x/abstract Wagner, J. (2012), "International Trade and Firm Performance: A survey of Empirical Studies since 2006," *Review of World Economics/Weltwirtschaftliches Archiv*, 148(2), 235–268. URL: <u>https://ideas.repec.org/a/spr/weltar/v148y2012i2p235-267.html</u>

Wooldridge, J. M. (2005), "Simple Solutions to the Initial Conditions Problem in Dynamic, Nonlinear Panel Data Models with Unobserved Heterogeneity," *Journal of Applied Econometrics*, 20, 39–54. URL: <u>http://onlinelibrary.wiley.com/doi/10.1002/jae.770/abstract</u>

# **Tables and Figures:**

## TABLE 1

## Number of Firms and Total Employment over 2003-2010

	Number of Firms	Number of Employee
2003	14,788	1232802
2004	16,446	1482741
2005	18,463	1717504
2006	19,536	1817297
2007	18,481	1874599
2008	17,926	1853687
2009	15,487	1631150
2010	21,089	1957774

## TABLE 2

## **Trade Participation and Employment Rates by Trade Status**

		P	ercentage of Firm	IS		Percentage of Employees				
	2003	2010	Same Status	Exit	Entry	2003	2010			
Non-Traders	40	38.23	16.7	72.3	111.7	14.3	19.6			
<b>Only-Exporters</b>	8.53	11.98	16.3	49.4	148.5	4.4	5.9			
<b>Only-Importers</b>	13.99	12.1	18.2	43.6	75.4	12.8	11.2			
Two-way Traders	37.5	37.69	53.2	33.6	70.8	68.5	63.4			
Total			30.5	51.8	94.5					

Notes: Columns 3,4 and 5 give percentage of firms according to 2003 values. The 3rd column gives those that had not changed status in 2010. Colums 4 and 5 show exit and entry of firms according to 2003.

		Start	Stop		
	2003	Trading	Trading	Switch	2010
Non-Traders	100	11.0	7.9		136.4
<b>Only-Exporters</b>	100	13.9	9.9	24.4	200.4
<b>Only-Importers</b>	100	9.3	9.2	29.0	123.3
Two-way Traders	100	5.1	2.8	10.4	143.3
Total	100				142.6

### **Transition of Firms Between Trading Categories (2003-2010)**

Notes: The Table 3 gives percentage of firms according to 2003 values. The first column reports the number of firms existing in each category in 2003. The next three columns report the switches of continuing firms in and out of each status The movements between non-traders and the three types of traders are reported in column 2 and 3, while in column 4 we report those traders that switch trading status.

### TABLE 4

### **Distribution of Exports Along the Extensive Margins (2003)**

			NC	E		
%Exporting Firms		15	610	1120	) 21+	Total
	1—5	46.4	6.9	3.9	0.9	58
NPE	6—10	8.8	4.7	3.2	1.3	18
NPE	1120	5.2	3.6	3.2	1.8	14
	21+	2.3	2.5	2.7	2.5	10
	Total	63	18	13	6	100
% Export Volume						
	1—5	4.8	3.8	3.9	2.5	15
NPE	6—10	1.9	2.1	4.6	5	14
NFE	1120	1.7	2.8	5.2	8.9	19
	21+	1.3	3	7	41.3	53
	Total	10	12	21	58	100

			NC	Ι		
%Importing Firms		15	610	1120	) 21+	Total
	1—5	49.8	1.6	0.1	0	51
NPI	6—10	12.5	6	0.5	0	19
INFI	1120	5.3	9.3	2.9	0	18
	21+	1.9	5.9	11.7	4.3	24
	Total	70	23	15	4	112
%Import Volume						
	1—5	3	0.5	0.1	0	4
NPI	6—10	1.7	2.2	0.3	0	4
INFI	1120	1.1	3.8	2.4	0	7
	21+	0.7	4.3	24.4	55.6	85
	Total	6	11	27	56	100

**Distribution of Imports Along the Extensive Margins (2003)** 

## TABLE 6

Firm Performance According to Trade Status (2003-2010)

	LP	TFP	Employee	Capint	Wage_L
Exporters	10.16	7.76	138.89	10.83	8.79
Importers	10.24	7.83	144.23	10.91	8.82
TW traders	10.29	7.87	164.06	10.96	8.86
Non-traders	9.49	7.17	48.93	9.97	8.51
<b>Only-Exporters</b>	9.67	7.35	47	10.34	8.53
Only-Importers	10.07	7.69	85.18	10.79	8.68

# Trade Premia Regressions (2003-2010)

		Sales			Employee	
	Pooled		Dynamic	Pooled		Dynamic
	Regression	FE	FE	Regression	FE	FE
Two-way trader dummy	1.715***	0.293***	0.181***	0.770***	0.153***	0.0847***
	(0.00799)	(0.00901)	(0.00793)	(0.00535)	(0.00575)	(0.00464)
Only-export dummy	0.458***	0.123***	0.0725***	0.0667***	0.0505***	0.0252***
	(0.00932)	(0.00835)	(0.00781)	(0.00533)	(0.00523)	(0.00452)
Only-import dummy	1.211***	0.169***	0.112***	0.408***	0.0891***	0.0539***
	(0.0101)	(0.00809)	(0.00735)	(0.00661)	(0.00513)	(0.00431)
<b>Observations</b>	126922	126922	98056	127171	127171	98319
R-squared	0.380	0.097	0.187	0.209	0.065	0.278
		LP			TFP	
	Pooled		Dynamic	Pooled		Dynamic
	Regression	FE	FE	Regression	FE	FE
Two-way trader dummy	0.647***	0.146***	0.129***	0.411***	0.131***	0.116***
	(0.00658)	(0.00976)	(0.0106)	(0.00679)	(0.0100)	(0.0108)
Only-export dummy	0.174***	0.0732***	0.0609***	0.0769***	0.0715***	0.0584***
	(0.00782)	(0.00976)	(0.0107)	(0.00834)	(0.0101)	(0.0109)
Only-import dummy	0.512***	0.0846***	0.0807***	0.314***	0.0773***	0.0720***
	(0.00813)	(0.00957)	(0.0103)	(0.00837)	(0.00988)	(0.0105)
<b>Observations</b>	115987	115987	87308	111645	111645	85446
R-squared	0.231	0.036	0.039	0.771	0.014	0.011
		Capint			Wage_L	
	Pooled		Dynamic	Pooled		Dynamic
	Regression	FE	FE	Regression	FE	FE
Two-way trader dummy	0.962***	0.0696***	0.0488***	0.163***	0.0343***	0.0248***
	(0.0120)	(0.00879)	(0.00572)	(0.00296)	(0.00350)	(0.00377)
Only-export dummy	0.375***	0.0218***	0.0172***	0.00692	0.00934***	0.00417
	(0.0151)	(0.00833)	(0.00573)	(0.0316)	(0.00339)	(0.00373)
Only-import dummy	0.792***	0.0538***	0.0416***	0.0994***	0.0184***	0.0171***
	(0.0147)	(0.00811)	(0.00553)	(0.00374)	(0.00332)	(0.00350)
Observations	119337	119337	94203	127171	127171	98343
R-squared	0.135	0.295	0.607	0.874	0.649	0.610

Notes: Reported are the estimated regression coefficients and the robust standard errors (in parentheses) from estimations of the dependent variables as number of employees (Employee), real manufacturing sales (Sales), labor productivity (LP), total factor productivity (TFP), capital intensity (Capint) and wages per employee (Wage\_L) at time t respectively. Asterisks denote significance levels (\*\*\*: p < 1%;\*\*: p < 5%; \*:p < 10%). All regressions include region, sector, foreign affiliation and year dummies as controls. LP, TFP, Capint and Wage\_L regressions also include logarithm of firms' number of employees as control, Dynamic FE regressions include lagged dependent variables. All dependent variables are in natural logarithms.

## **Ex-ante Performance Differentials of Trade Starters**

					Sa	les				
	( <i>t</i> -2)	(t-1)	(t-2)	( <i>t</i> -1)	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)
Non-trader that starts to export in t (dummy)	0.518*** (0.0323)	0.625*** (0.0323)								
Non-trader that starts to import in t (dummy)			0.844*** (0.0330)	0.945*** (0.0326)						
Non-trader that start to two-way trade (dummy)			× ,	· · · ·	0.895*** (0.0556)	0.959*** (0.0547)				
Importer that starts to export in t (dummy)					(0.0000)	(0102 17)	0.237*** (0.0405)	0.346*** (0.0384)		
Exporter that starts to import in t (dummy)									0.718*** (0.0378)	0.852*** (0.0366)
Observations	13220	17332	13428	17544	12657	16767	2686	3036	1914	2264
R-squared	0.066	0.066	0.104	0.102	0.070	0.068	0.057	0.068	0.104	0.106
					Emp	oloyee				
	( <i>t</i> -2)	(t-1)	(t-2)	( <i>t</i> -1)	( <i>t</i> -2)	(t-1)	( <i>t</i> -2)	( <i>t</i> -1)	(t-2)	( <i>t</i> -1)
Non-trader that starts to export in t (dummy)	0.136*** (0.0220)	0.206*** (0.0209)								
Non-trader that starts to import in t (dummy)			0.192*** (0.0300)	0.266*** (0.0284)						
Non-trader that start to two-way trade					0.196*** (0.0323)	0.268*** (0.0339)				
Importer that starts to export in t (dummy)					. ,	. ,	0.115*** (0.0169)	0.035** (0.0168)		
Exporter that starts to import in t (dummy)							. ,		0.189*** (0.0178)	0.143*** (0.0180)
Observations	13235	17369	13441	17583	12671	16801	2692	3040	1915	2264
R-squared	0.045	0.038	0.044	0.040	0.046	0.040	0.034	0.049	0.085	0.094

					L	Р				
	( <i>t</i> -2)	( <i>t</i> -1)	(t-2)	( <i>t</i> -1)	(t-2)	(t-1)	( <i>t</i> -2)	(t-1)	(t-2)	( <i>t</i> -1)
Non-trader that starts to export in t (dummy)	0.195*** (0.0278)	0.199*** (0.0272)								
Non-trader that starts to import in t (dummy)			0.258*** (0.0467)	0.321*** (0.0452)						
Non-trader that start to two-way trade			<b>`</b>	· · · ·	0.377*** (0.0281)	0.422*** (0.0265)				
Importer that starts to export in t (dummy)					(0.0200)	(	0.072** (0.0348)	0.107*** (0.0326)		
Exporter that starts to import in t (dummy)							(0.05 10)	(0.0320)	0.236*** (0.0363)	0.275*** (0.0317)
Observations	10678	13623	10867	13818	10142	13096	2581	2902	1818	2177
<u>R-squared</u>	0.076	0.076	0.089	0.094	0.080	0.081	0.067	0.086	0.083	0.120
					TI	<b>P</b>				
	( <i>t</i> -2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)
Non-trader that starts to export in t (dummy)	0.246*** (0.0596)	0.253*** (0.0599)								
Non-trader that starts to import in t (dummy)			0.274***	0.311***						
			(0.0467)	(0.0452)						
Non-trader that start to two-way trade					0.319***	0.445***				
Importer that starts to export in t (dummy)					(0.0963)	(0.0901)	0.065** (0.0321)	0.101***		
Exporter that starts to import in t (dummy)							(0.0321)	(0.0255)	0.191** (0.0748)	0.246*** (0.0689)
Observations	10073	12667	10266	12869	9554	12160	2549	2854	1784	2119
R-squared	0.057	0.050	0.058	0.052	0.059	0.053	0.084	0.092	0.112	0.117

# TABLE 8 (Cont'd)

					Capi	int				
	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)
Non-trader that starts to export in t (dummy)	0.420*** (0.0558)	0.456*** (0.0526)								
Non-trader that starts to import in t (dummy)			0.692*** (0.0935)	0.768*** (0.0860)						
Non-trader that start to two-way trade			. ,		0.836*** (0.0505)	0.893*** (0.0473)				
Importer that starts to export in t (dummy)					()	(0.00.000)	0.209*** (0.0652)	0.193*** (0.0599)		
Exporter that starts to import in t (dummy)							(010002)	(0100333)	0.560*** (0.0729)	0.530*** (0.0649)
Observations	11630	14778	11839	14996	11087	14235	2654	2984	1877	2199
R-squared	0.043	0.045	0.057	0.057	0.043	0.044	0.042	0.044	0.093	0.080
					Wage	e_L				
	( <i>t</i> -2)	(t-1)	(t-2)	( <i>t</i> -1)	(t-2)	(t-1)	(t-2)	(t-1)	(t-2)	(t-1)
Non-trader that starts to export in t (dummy)	0.0236 (0.0113)	0.0146 (0.0157)								
Non-trader that starts to import in t (dummy)			0.0422*** (0.0142)	0.0460*** (0.0136)						
Non-trader that start to two-way trade (dummy)					0.0541*** (0.0117)	0.0908*** (0.0106)				
Importer that starts to export in t (dummy)							0.0225 (0.0168)	0.00397 (0.0111)		
Exporter that starts to import in t (dummy)									0.0235 (0.0187)	0.0409** (0.0186)
Observations	13235	17369	13441	17583	12671	16801	2692	3040	1915	2264
<i>R-squared</i>	0.713	0.698	0.714	0.704	0.714	0.700	0.798	0.801	0.756	0.755

### TABLE 8 (Cont'd)

Notes: Reported are the estimated regression coefficients and the robust standard errors (in parentheses) from estimations of the dependent variables as number of employees (Employee), real manufacturing sales (Sales), labor productivity (LP), total factor productivity (TFP), capital intensity (Capint) and wages per employee (Wage\_L) at time t-2 and t-1 respectively. Asterisks denote significance levels (\*\*\*: p < 1%; \*\*: p<5%; \*:p<10%). All regressions include region, sector, foreign affiliation and year dummies as controls. LP, TFP, Capint and Wage\_L regressions also include logarithm of firms' number of employees as control, All dependent variables are in natural logarithms.

	Pan	el A(without to	uriffs)	Pa	nel B(with tar	iffs)
	Only- exporter	Only- importer	Two-way trader	Only- exporter	Only- importer	Two-way trader
Only-export dummy(t-1)	0.921***			0.878***		
	(0.0269)			(0.0260)		
Only-import dummy(t-1)		0.959***			0.949***	
		(0.0225)			(0.0223)	
Two-way trader dummy(t-1)			1.072***			1.055***
			(0.0217)			(0.0211)
Exporter dummy(t-1)						
Importer dummy(t-1)						
Employee(t-1)	0.0889**	0.0925**	0.112**	0.0855**	0.0997**	0.115***
	(0.0371)	(0.0475)	(0.0451)	(0.0442)	(0.0464)	(0.044)
TFP(t-1)	0.0215***	0.0348***	0.0416***	0.0268***	0.0335***	0.0483***
	(0.0067)	(0.0048)	(0.0138)	(0.005)	(0.0043)	(0.0134)
Wage_L(t-1)	0.0100	0.0203**	0.0603*	0.0109	0.0174***	0.0604**
	(0.0449)	(0.0083)	(0.0364)	(0.0424)	(0.0057)	(0.0354)
Observations	85412	85412	85412	85412	85412	85412

### **Dynamic Panel Probit Regressions**

Notes: The table reports dynamic panel probit regressions. (t - 1) indicates that the variable is lagged. Reported are the estimated regression coefficients and the robust standard errors (in parentheses) from estimations of the dependent variables as binary outcome variables of being an only exporter, only importer and two way trader respectively. Asterisks denote significance levels (\*\*\*: p < 1%;\*\*: p<5%; \*:p<10%). All regressions include means of the continuous explanatory variables and initial values of the dependent variables as well as region, sector, foreign affiliation and year dummies as controls.

## Dynamic Probit Regressions w.r.to BEC Classification

			WITHOUT	TARIFFS	5				WITH T	ARIFFS		
	Capital Exporter Only	Intermediate Exporter Only	Consumption Exporter Only	Capital Importer Only	Intermediate Importer Only	Consumption Importer Only	Capital Exporter Only	Intermediate Exporter Only	Consumption Exporter Only	Capital Importer Only	Intermediate Importer Only	Consumption Importer Only
Capital Exporter Only (t-1)	0.933***						0.919***					
	(0.0213)						(0.0387)					
Intermediate Exporter Only (t-1)		0.925***						0.914***				
		(0.0782)						(0.0248)				
Consumption Exporter Only (t-1)			0.873***						0.820***			
			(0.0406)						(0.0396)			
Capital Importer Only (t-1)				0.992***						0.974***		
				(0.0411)						(0.0387)		
Intermediate Importer Only (t-1)					0.961***						0.923***	
					(0.0257)						(0.0277)	
Consumption Importer Only (t-1)						0.874***						0.831***
						(0.0438)						(0.0737)
Observations	82869	83105	83278	82696	83278	83278	82869	83105	83278	82696	83278	83278

Notes: The table reports dynamic panel probit regressions. (t - l) indicates that the variable is lagged. Reported are the estimated regression coefficients and the robust standard errors (in parentheses) from estimations of the dependent variables as binary outcome variables of being an only exporter, only importer and two way trader respectively. Asterisks denote significance levels (\*\*\*: p < 1%;\*\*: p < 1%;\*\*: p < 5%; \*:p < 10%). All regressions include means of the continuous explanatory variables and initial values of the dependent variables as well as region, sector, foreign affiliation and year dummies as controls.

Mean Productivity of Two-way Traders with respect to Extensive Margins (2003-2010)

	NPE		NPI		NCE		NCI	
	TFP	LP	TFP	LP	TFP	LP	TFP	LP
15	7.62	10.04	7.55	9.97	7.68	10.05	7.59	10.02
610	7.75	10.21	7.85	10.23	7.84	10.24	8.08	10.43
1120	7.96	10.29	8.06	10.37	7.84	10.42	8.28	10.68
20+	8.23	10.48	8.25	10.74	8.09	10.7	8.52	11.07

### TABLE 12

### Trade Premia Along Extensive Margins (2003-2010)

	Sales	Employee	LP	TFP	Capint	Wage_L
NPE	0.0262***	0.00812**	0.0108*	0.0104*	0.00159	0.00199
	(0.00456)	(0.00362)	(0.00609)	(0.00622)	(0.00471)	(0.00244)
NPI	0.101***	0.0784***	0.0548***	0.0433***	0.0526***	0.0222***
	(0.00594)	(0.00432)	(0.00793)	(0.00812)	(0.00694)	(0.00333)
NCE	0.0824***	0.0590***	0.0518***	0.0503***	0.00828	0.0174***
	(0.00557)	(0.00465)	(0.00703)	(0.00716)	(0.00677)	(0.00396)
NCI	0.125***	0.0777***	0.0797***	0.0744***	0.0143*	0.0190***
	(0.00727)	(0.00559)	(0.00957)	(0.00978)	(0.00810)	(0.00309)
<b>Observations</b>	52872	52906	51207	50541	52128	52906
<b>R-squared</b>	0.194	0.139	0.064	0.026	0.308	0.639

Notes: Reported are the estimated regression coefficients and the robust standard errors (in parentheses) from estimations of the dependent variables as number of employees (Employee), real manufacturing sales (Sales), labor productivity (LP), total factor productivity (TFP), capital intensity (Capint) and wages per employee (Wage\_L) at time t respectively. Asterisks denote significance levels (\*\*\*: p < 1%;\*\*: p < 5%; \*:p < 10%). All regressions include region, sector, foreign affiliation and year dummies as controls. LP, TFP, Capint and Wage\_L regressions also include logarithm of firms' number of employees as control, All dependent variables are in natural logarithms.



Please note:

You are most sincerely encouraged to participate in the open assessment of this discussion paper. You can do so by either recommending the paper or by posting your comments.

Please go to:

http://www.economics-ejournal.org/economics/discussionpapers/2015-17

The Editor